Author's response to reviews

Title: Malnutrition and bilateral central retinal vein occlusion in a young woman: a case report

Authors:

Mark Taubert (mtaubert@doctors.org.uk)
Tim C Dowd (TIMDREDAED@aol.com)
Angela Wood (Angela.Wood@stees.nhs.uk)

Version: 6 Date: 3 March 2008

Author's response to reviews:

Abstract. The abstract must be structured into the following sections:
Introduction
Case presentation
Conclusions

Please ensure the abstract is identical in both the manuscript and submission system.
Done

References in the manuscript text at the end of a sentence must be placed before the full stop.
Done

Authors' contributions - Please indicate that all authors read and approved the final manuscript.
Done

Consent. Please expand the consent section to read as follows: Written informed consent was obtained from the patient for publication of this Case report. A copy of the written consent is available for review by the Editor-in-Chief of this journal.
Done.

Reference list - The reference list should contain all authors' names, if there are 30 or below, the term 'et al.' should not be used. If there are more than 30 names a minimum of 30 should be listed, and then et al can be used.
Done